Author's response to reviews

Title: Implementation of tuberculosis infection control in health facilities in Mukono and Wakiso districts, Uganda

Authors:

Esther EB Buregyeya (eburegyeya@musph.ac.ug)
Suzanne SV Verver (VerverS@kncvtbc.nl)
Bart BC Criel (BCriel@itg.be)
Fred FN Nuwaha (fnuwaha@musph.ac.ug)
Robert RC Colebunders (bcoleb@itg.be)
Rhoda RW Wanyenze (rwanyenze@musphcdc.ac.ug)
Joan N Kalyango (nakayaga2001@yahoo.com)
Achilles AK Katamba (akatamba@yahoo.com)
Ellen MH Mitchell (MitchellE@kncvtbc.nl)

Version: 4 Date: 15 May 2013

Author's response to reviews: see over
15/05/2013

The Chief Editor
BMC Infectious Diseases Journal,

Dear Sir/Madam,

RE: Implementation of tuberculosis infection control in health facilities in Mukono and Wakiso districts, Uganda

On behalf of my co-authors, I take this opportunity to re-submit to your Journal of BMC Infectious Diseases, for publication of an original article titled: “Implementation of tuberculosis infection control in health facilities in Mukono and Wakiso districts, Uganda.” I have addressed the reviewers, comments.

I will be very grateful if I receive your favourable consideration.

Yours sincerely,

Esther Buregyeya
Lecturer,
Department of Disease Control & Environmental Health
Makerere University School of Public Health
P.O.Box 7062, Kampala, Uganda.
E-mail: eburegyeya@musph.ac.ug